Author's response to reviews

Title: Epidemiology analysis of injury in Shandong Province, China

Authors:

Jixiang Ma (majixiang@gmail.com)
Xiaolei Guo (guoxiaolei@126.com)
Aiqiang Xu (aqxuepi@163.com)
Jiyu Zhang (zhangjiyu008@hotmail.com)
Chongqi Jia (jiachongqi@sdu.edu.cn)

Version: 8 Date: 14 March 2008

Author's response to reviews: see over
Dear Editor,

I have revised the manuscript according to the requirements. All revisions were marked in blue. Below are my revisions to the comments by Professor Mei Wen.

1. The main description is on epidemic features. Suggest taking health economic analysis off from the topic, since there was only one short paragraph on it. If included, there needs more description and interpretation from the economics side.

Revised according to comments.

2. The goodness of fit test in Table 5 showed a poor fit, indicating model prediction is significantly different from observed values, some discussion may be needed here, and the relationship might be non-linear.

Comments have been added in the limitation of the study. I am sorry the P value should be less than 0.05.

3. Please check the inconsistency of the numbers in the description and the tables.

Modified according to suggestion.